Author’s response to reviews

Title: Meaning in life and health care use: findings from a nationally representative study of older adults in Germany

Authors:

André Hajek (a.hajek@uke.de)
Hans-Helmut König (h.koenig@uke.de)

Version: 3 Date: 12 Dec 2019

Author’s response to reviews:

1. We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap exists throughout your manuscript.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.

--&gt; Thank you very much for your comment. Please see the e-mail correspondence with Mrs Eloisa Nolasco (“In line with your concerns, your justification for the overlap is fine.”)

2. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

--&gt; Thank you for your comment. We added such a list of abbreviations (before the Declarations section)
3. Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

--&gt; We have done it accordingly.